Author's response to reviews

Title: E-selectin gene polymorphisms are associated with essential hypertension: a case-control pilot study in Chinese

Authors:

Zuoguang Wang (wzg70@hotmail.com)
Ya Liu (yaya021@sohu.com)
Jieling Liu (jily1979@263.net)
Kuo Liu (kuoliu010@sohu.com)
Shaoyun Wen (wzg70@yahoo.com)
Jie Wen (jiew1001@sohu.com)
Zhaosu Wu (wuzhaosu123@263.net)

Version: 9 Date: 6 August 2010

Author's response to reviews: see over
Dear Editor

We are now resubmitting a revised manuscript by Zuoguang Wang, et al, entitled “(MS: 1617310499373018) E-selectin gene polymorphisms are associated with essential hypertension: a case-control pilot study in a Chinese population” for the possible publication in BMC Medical Genetics.

Following the comments of the two reviewers and the editorial office, we have copyedited the paper to improve the style of written English by using a professional copyediting service from International Science Editing (http://www.internationalscienceediting.com/) as the editor recommended. We expect the revised manuscript can meet the requirement. If anything about the manuscript is necessary, please contact us as soon as possible.

Yours, sincerely

Shaojun Wen, M.D.